

Dear Editors:

How are you! The author has invited a formal service company to polish the English language as required and issued a certificate. Please note: English-speaking experts have made some language changes to the original text, but they have not changed the basic content; The references are all perfected and supplemented by the author.

In addition, for the convenience of reading, all modified logos are no longer displayed.

With best regards

Professor Hua Bai MD and Ph.D

Guizhou Medical University in China